Title: A large community outbreak of waterborne giardiasis - delayed detection in a non-endemic urban area

Authors:

Karin Nygard (kany@fhi.no)
Barbara Schimmer (basc@fhi.no)
Oystein Sobstad (oysteinsobstad@gmail.com)
Anna Karen Walde (Anna.Karen.Walde@mattilsynet.no)
Ingvar Tveit (Ingvar.Tveit@bergen.kommune.no)
Nina Langeland (nina.langeland@helse-bergen.no)
Trygve Hausken (trygve.hausken@helse-bergen.no)
Preben Aavitsland (praa@fhi.no)

Version: 5 Date: 10 May 2006

Author's response to reviews: see over
To the Central Editorial Production Team,
BMC Public Health

Oslo, 2006-05-10

Manuscript titled: A large community outbreak of waterborne giardiasis- delayed detection in a non-endemic urban area

Formatting changes requested:
Affiliations - Each affiliation should be written in full and be provided in the following format: department, institute, city, country. Can we also ask you to link each author to their corresponding affiliation with the use of superscript numbers.

Corrected
Authors contributions - Please change to Authors' contributions.

Changed
Figures - It is important for the final layout of the manuscript that the figures are cropped as closely as possible to minimise white space around the image.

Changed

I have made the changes according to the guidelines. Please do not hesitate to contact me again if further changes are needed.

Best wishes,
Karin Nygård

Norwegian Institute of Public Health
Ph: +47 22 04 24 91
e-mail: kany@fhi.no